Author’s response to reviews

Title: Predictors of Burnout, Work Engagement and Nurse Reported Job Outcomes and Quality of Care: a Mixed Method Study

Authors:

Peter Van Bogaert (peter.vanbogaert@uantwerpen.be)
Lieve Peremans (lieve.peremans@uantwerpen.be)
Danny Van heusden (danny.vanheusden@uantwerpen.be)
Martijn Verspuy (martijn.verspuy@uantwerpen.be)
Veronika Kureckova (veronikakureckova@gmail.com)
Zoe Van de Cruys (zoe.vandercruys@uza.be)
Erik Franck (erik.franck@uantwerpen.be)

Version: 2 Date: 03 Nov 2016

Author’s response to reviews:

Dear Editor,

We have adjusted (see in red) the abstract and references as the reviewers suggested.

Thank you very much to accept our paper for publication.

The authors